Author’s response to reviews

Title: Motivational Interviewing as an intervention to increase adolescent self-efficacy and promote weight loss: Methodology and design

Authors:

Beverly Walpole (bwalpole@uoguelph.ca)
Elizabeth Dettmer (elizabeth.dettmer@sickkids.ca)
Barbara Morrongiello (bmorrong@uoguelph.ca)
Brian McCrindle (brian.mccrindle@sickkids.ca)
Jill Hamilton (jill.hamilton@sickkids.ca)

Version: 6 Date: 30 May 2011

Author’s response to reviews:

The figures have been reordered and are now in consecutive citation order.

In the abstract, a change from the heading "Conclusion" to "Discussion".

In the body of the manuscript, a change in the heading "Anticipated results and discussion" to "Discussion".

The figures have been edited and cropped to remove titles and minimize white space around the image.